Author's Response To Reviewer Comments

Clo<u>s</u>e

We believe these revisions will address all the concerns raised by you and the reviewers. We apologies for the misunderstanding with respect to the ROC curve and AUC values. We have added these in the legend for each dataset. We have also added ISGPRE to biotools and registered ISGPRE with scicrunch.com and received the RRID:SCR_022730, however it does not yet appear when browsing the scicrunch dashboard. We have also made a few additional clarifications to the text, in particular, after much discussion, we decided to change the 'noisy' feature terminology to poorly performing feature (this was originally called disruptive feature).

Clo<u>s</u>e